Author’s response to reviews

Title: Adult T-type lymphoblastic lymphoma presenting as hypercalcaemic crisis and aplastic anaemia-An unusual presentation of a rare disease: a case report

Authors:

Mickael Essouma (essoumam@gmail.com)
Dorothée Soh (doryht07@yahoo.fr)
Mazou Temgoua (neurotemgoua@yahoo.fr)
Ronald Gobina (ronaldgob@yahoo.fr)
Aristide Nono (notarise@yahoo.fr)
Etienne Olivier Atenguena (Atenguenatien2002@yahoo.com)
Mahamat Maimouna (m_mahamat@yahoo.fr)
Gloria Ashuntantang (maglo09@hotmail.com)

Version: 1 Date: 05 Aug 2019

Author’s response to reviews:

To

The Editorial Office

Journal of Medical Case Reports

Title: Adult T-type lymphoblastic lymphoma presenting as hypercalcaemic crisis and aplastic anaemia-An unusual presentation of a rare disease: a case report

Dear Editor,

We have revised our manuscript, JMCR-D-19-00334. We thank you and the reviewer for your constructive comments and hope the manuscript is now suitable for publication accordingly. We give a point-by-point response to the reviewer’s comments below. Revisions appear in red colour in the revised manuscript with tracked changes.
Comments

1- Add a paragraph at the end of the introduction that explains why this case report is presented (what is unique and adds to the medical knowledge)

2- More information is needed:

The case report should include past medical, social, environmental, family and employment history. Where there any risk factors or toxic exposure that could have lead to aplastic anaemia?

What medications was the patient receiving prior to diagnosis? Did the patient smoke, and/or consume alcohol?

Give detailed physical and neurological examination on admission.

Give the doses of all medications that were given and their doses.

Was an autopsy performed, and what were the findings?

3-Discussion-add a paragraph at the beginning of the discussion that summarizes the case and describes what is unique in this case compared to what is available in the literature.

Responses

1- We revised the Background section accordingly

2- More informations were added regarding the past medical history, doses of durgs, and physical exam as you can read in the Case presentation section. No autopsy was done, and this was acknowledged as a limitation in the Discussion section

3- We added a paragraph at the beginning of the discussion accordingly.